Author’s response to reviews

Title: Cognitive behavioural therapy versus supportive therapy for persistent positive symptoms in psychotic disorders: The POSITIVE Study, a multicenter, prospective, single-blind, randomised controlled clinical trial

Authors:

Stefan Klingberg (stefan.klingberg@med.uni-tuebingen.de)
Andreas Wittorf (andreas.wittorf@med.uni-tuebingen.de)
Christoph Meisner (christoph.meisner@med.uni-tuebingen.de)
Wolfgang Wölwer (wolfgang.woelwer@uni-duesseldorf.de)
Georg Wiedemann (georg.wiedemann@klinikum-fulda.de)
Jutta Herrlich (jutta.herrlich@kgu.de)
Andreas Bechdolf (andreas.bechdolf@uk-koeln.de)
Bernhard W Mueller (bernhard.mueller@uni-due.de)
Gudrun Sartory (sartory@uni-wuppertal.de)
Michael Wagner (michael.wagner@ukb.uni-bonn.de)
Tilo Kircher (kircher@med.uni-marburg.de)
Hans-Helmut Koenig (h.koenig@uke.uni-hamburg.de)
Corinna Engel (corinna.engel@med.uni-tuebingen.de)
Gerhard Buchkremer (helma.buxmann@med.uni-tuebingen.de)

Version: 3 Date: 11 November 2010

Author’s response to reviews: see over
Re-Submission to “Trials”: Cognitive behavioural therapy versus supportive therapy for persistent positive symptoms in psychotic disorders: The POSITIVE Study, a multicenter, prospective, single-blind, randomised controlled clinical trial

Dear Editors:

We would like to resubmit the abovementioned revised manuscript for reconsideration for publication in Trials.

We are happy that the paper found your interest and thank you very much for your and the reviewers' comments. We carefully reconsidered the critical aspects and think that we could address the comments sufficiently.

- The trial registration number was indeed incorrect, the last character was missing. We included the correct ISRCTN in the manuscript.
- We reworded the title so it does not include full stops. The title reads now: “Cognitive behavioural therapy versus supportive therapy for persistent positive symptoms in psychotic disorders: The POSITIVE Study, a multicenter, prospective, single-blind, randomised controlled clinical trial”

We look forward to hearing from you soon,

Sincerely

Prof. Dr. Stefan Klingberg